Author's response to reviews

Title: Combining evidence and values in priority setting: testing the balance sheet method in a low-income country

Authors:

Emmanuel Makundi (emakundi@nimr.or.tz)
Lydia Kapiriri (Lydia.Kapiriri@utoronto.ca)

Version: 4 Date: 23 August 2007

Author's response to reviews:

Dear Editor

We have now revised the text according to the reviewer’s comments. These are as follows:

First, we are now consistent when we refer to the WHO-CHOICE database (we have deleted the term ‘the WHO generalised cost-effectiveness database’).

Second, we have structured our abstract according to the guidelines of the journal. The manuscript now also conforms to the journal style.

We apologize for the delay in our response.

Yours, sincerely

Prof. Ole Frithjof Norheim